\begin{table}%t1
%\centering
\par
\caption{\label{tab:models}Parameter names used throughout the paper.}
\begin{tabular}{lll}
\hline\hline
Parameter name & Parameter description & References\\
\hline
{\tt geneva05} & stellar evolution with rotation & \citet{Meynet2005,Palacios}\\
{\tt geneva05alt} & stellar evolution with rotation & \citet{Meynet2005}\\
{\tt geneva97} & stellar evolution & \citet{MaederM1994,Meynet1997}\\
{\tt LC06} & stellar evolution & \citet{Limongi}\\
{\tt wind08} & wind velocities & \citet{Lamers1995,Niedzielski}\\
{\tt wind00} & wind velocities & \citet{Howarth,Prinja}\\
{\tt yieldsLC06} & supernova nucleosynthesis yields & \citet{Limongi}\\
{\tt yieldsWW95} & supernova nucleosynthesis yields & \citet{Woosley1995,Cervino2000}\\
{\tt atmosMS} & stellar atmospheres & \citet{Kurucz,MartinsF,Smith2005}\\
{\tt atmosSmith} & stellar atmospheres & \citet{Kurucz,Smith2005}\\
{\tt atmosOLD} & stellar atmospheres & \citet{Kurucz,Schmutz,Schaerer}\\
\hline
\end{tabular}
\end{table}